Author's response to reviews

Title: Clinical predictors of statins prescription in acute ischemic stroke patients: findings from the Lombardia Stroke Registry

Authors:

Isabella Canavero (isabellacanavero@libero.it)
Anna Cavallini (anna.cavallini@mondino.it)
Patrizia Perrone (patrizia.perrone@ao-legnano.it)
Mauro Magoni (mago1959@hotmail.com)
Lucia Sacchi (lucia.sacchi@unipv.it)
Silvana Quaglini (silvana.quaglini@gmail.com)
Giordano Lanzola (giordano.lanzola@unipv.it)
Giuseppe Micieli (giuseppe.micieli@mondino.it)

Version: 2 Date: 9 July 2013

Author's response to reviews: see over
Dear Journal Editorial Office,

Apologize about the absence of some important features of our manuscript! Here the answers to Your questions are reported:

1. Ethics: Our study has been performed under the approval of the Ethical Committees of all the Institutes participating to Lombardia Stroke Registry network. You can find the extensive list of all the Institutes in the “Acknowledgements” section.

2. Competing interests: We have no competing interests, neither financial nor non-financial, to declare.

   ? In the past five years have you received reimbursements, fees, funding, or salary from an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? Is such an organization financing this manuscript (including the article-processing charge)? If so, please specify. 
   **No.**

   ? Do you hold any stocks or shares in an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? If so, please specify. **No.**

   ? Do you hold or are you currently applying for any patents relating to the content of the manuscript? Have you received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript? If so, please specify. **No.**

   ? Do you have any other financial competing interests? If so, please specify. **No.**

   ? Are there any non-financial competing interests (political, personal, religious, ideological, academic, intellectual, commercial or any other) to declare in relation to this manuscript? If so, please specify. **No.**

3. Authors' information: After “Competing interests” section, we have now specified the role that every single Author has played in the paper.

I hope that our paper could draw Your attention and fulfill the criteria to be published in Your Journal.

With my best regards,

Dr. Isabella Canavero